Author's response to reviews

Title: Effectiveness of a stepped primary care smoking cessation intervention (ISTAPS): design of a cluster randomised trial

Authors:

Carmen Cabezas (carmen.cabezas@gencat.cat)
Carlos Martin (Carlos.Martin@uab.es)
Silvia Granollers (sgranollers.cp.ics@gencat.cat)
Concepció Morera (concepcio.morera@pgirona.scs.es)
Josep Lluis Ballve (18325jbm@comb.es)
Elvira Zarza (ezarza@teleline.es)
Jordi Blade (jblade.tarte.ics@gencat.net)
Margarida Borras (mborras.tarte.ics@gencat.net)
Antoni Serra (clarinetcustom@telefonica.net)
Diana Puente (dpuente@idiapjgol.org)
Intervenció Sistemática Tabaquismo en Atenció Primaria Salut (ISTAPS) (carmen.cabezas@gencat.cat)

Version: 2 Date: 27 January 2009

Author's response to reviews:

1- Figures (1 and 2) have been removed from the manuscript and added at the end of the document after the references.
2- Authors group corrected and the name of the group has been added as an author. And the list of all participants included in the acknowledgments section.
3- Reference 4 corrected.
4- Reference 6 and 17 were already complete.
5- Title and abstract corrected.
6- E-mails changed.
7- Superscript on the title page added.
8- Headings sentence case case corrected.
9- Underlining removed and replace by bold.
10- Conclusions added.
11- Sentence in author's contributions added.
12- Removed the word Financing from Acknowledgments section.